Author’s response to reviews

Title: Being unvaccinated and having a contact history increased the risk of Measles Infection during an Outbreak: A finding from a Measles Outbreak Investigation in Rural district of Ethiopia

Authors:

Abadi Girmay (abadigirmay233@yahoo.com)
Abel Fekadu Dadi (Fekten@yahoo.com)

Version: 1 Date: 15 Feb 2019

Author’s response to reviews:

The response to editor and reviewers comment has been attached as a supplementary materials